Author's response to reviews

Title: A Knowledge Translation Collaborative to Improve the Use of Therapeutic Hypothermia in Post-Cardiac Arrest Patients: Study Protocol for a Stepped Wedge Cluster Randomized Trial

Authors:

Katie N. Dainty (kndainty@gmail.com)
Damon C. Scales (damon.scales@sunnybrook.ca)
Steven C. Brooks (brooksst@smh.ca)
Dale M. Needham (dale.needham@jhmi.edu)
Paul Dorian (dorianp@smh.ca)
Niall Ferguson (niall.ferguson@uhn.on.ca)
Gordon Rubenfeld (gordon.rubenfeld@sunnybrook.ca)
Randy Wax (randy.wax@gmail.com)
Merrick Zwarenstein (merrick.zwarenstein@ices.on.ca)
Kevin Thorpe (kevin.thorpe@utoronto.ca)
Laurie J Morrison (morrisonL@smh.ca)

Version: 5  Date: 12 January 2011

Author's response to reviews: see over
January 12, 2011

Chair, Editorial Board  
Implementation Science  
BioMed Central Ltd  
Floor 6, 236 Gray's Inn Road  
London WC1X 8HB  United Kingdom

RE: Revised Manuscript Submission

Dear Editorial Board,

Please accept the revised study protocol manuscript entitled “A Knowledge Translation Collaborative to Improve the Use of Therapeutic Hypothermia in Post-Cardiac Arrest Patients: Protocol for a Stepped Wedge Randomized Trial” as uploaded to the website.

We have corrected all of the formatting changes you have requested and made a final review of the manuscript. We approve the manuscript as submitted.

We are very excited to have the opportunity to publish in your exceptional journal.

Sincerely,

Katie Dainty, PhD  
Lead Author  
Post-Doctoral Research Fellow

Laurie Morrison, MD, MSc, FRCPC  
Senior Author, Primary Investigator  
Director, RESCU